Author's response to reviews

Title: The facilitating factors and barriers encountered in the adoption of a humanized birth care approach in a highly specialized university affiliated hospital

Authors:

Roxana Behruzi (roxana.behruzi@umontreal.ca)
Marie Hatem (marie.hatem@umontreal.ca)
Lise Goulet (lise.goulet@umontreal.ca)
William Fraser (william.fraser@umontreal.ca)

Version: 3 Date: 5 November 2011

Author's response to reviews: see over
November 5, 2011

Dear Editor

The authors thank you for your interest to the paper. Please find the corrections regarding to the second reviewer’s comments in following:

pg 3: “only” was added to the sentence: "obstetricians and family physicians who provide only antenatal care..."

pg 5: “other” was replaced by “majority” in the following sentence: “there were the majority of women who were cared for, at the same hospital, but did not suffer from complications described as being at high-obstetric-risk”.

Pg 16 “the term Valorization of” was replaced by “valuing of”

pg 24 “excited about” was replaced by “proud about”

The corrections regarding to the editor’s comments:

1- A title page has provided to the first page of the main manuscript.
2- The ‘Competing interests' section was added between the conclusions and authors' contributions
3- The authors' contributions section was added before the acknowledgements and reference list
4- The response to reviewers document was removed

We hope the corrections are satisfactory. Please don’t hesitate to contact me if there is anything else we can help with the paper.

I look forward to hearing from you.

Sincerely

Roxana Behruzi, PhD
Postdoctoral Fellow

Address
Department of Family Medicine
McGill University
115 Pine Avenue West
Montreal, Qc, H2W 1S4, Canada
Email: roksana.behruzi@mail.mcgill.ca